Author’s response to reviews

Title: Mapping evidence on the distribution of Human Papillomavirus related cancers in sub-Saharan Africa: Scoping Review Protocol

Authors:

Bridget K.M Lekoane (lekoane.bridgette@gmail.com)

Tivani P. Mashamba-Thompson (Mashamba-Thompson@ukzn.ac.za)

Themba G Ginindza (Ginindza@ukzn.ac.za)

Version: 4 Date: 01 Nov 2017

Author’s response to reviews:

RESPONSES TO REVIEWERS

REFERENCE: SYSR-D-17-00134

MANUSCRIPT TITLE: Mapping the evidence on the distribution of Human Papillomavirus related cancers in sub-Saharan Africa: Scoping Review

Comment 1

I would like the authors to make one change throughout the manuscript. Everywhere mention made of “systematic scoping review” should be replaced with “scoping review”

Response

Changes were made on the following pages and line

Page 1, Line 2

Page 2, line 18

Page 3, line 63

Page 4, line 71

Page 4, line 78

Page 9, line 164